

## **Author's response to reviews**

**Title:** Breastfeeding cessation and symptoms of anxiety and depression: a longitudinal cohort study

### **Authors:**

Eivind Ystrom ([eivind.ystrom@fhi.no](mailto:eivind.ystrom@fhi.no))

**Version:** 2 **Date:** 7 February 2011

### **Author's response to reviews:**

Dear Editor,

On January 26 I was requested to seek further assistance with the language of my manuscript "Breastfeeding cessation and symptoms of anxiety and depression: a longitudinal cohort study". The manuscript has now been amended by a professional scientific copyeditor.

Yours sincerely,

Eivind Ystrom